Author's response to reviews

Title: Application of Ni-Ti Alloy connector for the treatment of comminuted coronal plane supracondylar-condylar femoral fractures

Authors:

Yuntong Zhang (mone1030@126.com)
Xue Zhao (mone1030@sina.cn)
Yang Tang (mone1030@sina.com)
Chuncai Zhang (mone1030@163.com)
Shuogui Xu (zone1030@gmail.com)
Yang Xie (mone1030@yeah.net)

Version: 2 Date: 1 June 2013

Author's response to reviews: see over
Reviewer's report
Title: Application of Ni-Ti Alloy connector for the treatment of comminuted coronal plane supracondylar-condylar femoral fractures: a retrospective review of 21 patients
Version: 1 Date: 6/1/2013
Reviewer: Yuntong Zhang

Reviewer's report:

1.) Ethics statement: Research involving human subjects (including human material or human data) that is reported in the manuscript must have been performed with the approval of an appropriate ethics committee. Research carried out on humans must be in compliance with the Helsinki Declaration (http://www.wma.net/en/30publications/10policies/b3/index.html). A statement to this effect must appear in the Methods section of the manuscript, including the name of the body which gave approval, with a reference number where appropriate.

Ethics statement has been reported in the Methods section with highlighted text

2.) Kindly clarify whether patients provided written informed consent for the publication of individual clinical details (Please note that if this consent was not obtained, then table 1 must be removed)

written informed consent has been clarified in the Methods section with highlighted text

3.) Please add a "Title page" section to your manuscript. (Kindly note that this should be within the main manuscript document and not uploaded as additional file);

"Title page" section has been added in the first page of manuscript

4.) Please add an "Acknowledgements" section;

"Acknowledgements" section has been added at the end of main manuscript